Author's response to reviews

Title: Governance of evolving response to Non-communicable diseases in low-and middle- income countries: current status, risks

Authors:

Manju Rani (ranim@wpro.who.int)
Sharmin Nusrat (night_sky7@hotmail.com)
Hawken Laura (hawkenl@wpro.who.int)

Version: 2 Date: 24 May 2012

Author's response to reviews: see over
To

Editor
BMC Public Health

Subject: Submission of an original manuscript

Sir,

I am pleased to resubmit the manuscript titled “Governance of evolving response to Non-communicable diseases in low-and middle-income countries: current status, risks and options” on behalf of myself and other authors.

Following changes have been made in the revised version:

1) A separate results section has been added in the abstract.
2) A section on declaration of competing interests has been included after the ‘conclusion’ section.
3) Authors information: A section on authors contribution has been included after competing interest section.
4) Data availability: Public availability of the document reviewed and the other data used in the analysis has been now specifically mentioned in the ‘methods’ section.

Please do let us know if you need any other information. We look forward to a favourable response.

Best regards

Yours sincerely

Manju rani

(Dr. Manju Rani)